Author's response to reviews

Title: SPECT myocardial perfusion imaging as an adjunct to coronary calcium score for the detection of hemodynamically significant coronary artery stenosis

Authors:

Franz von Ziegler (franz.ziegler@med.uni-muenchen.de)
Michaela Brendel (michaela.brendel@yahoo.de)
Christopher Uebleis (christopher.uebleis@med.uni-muenchen.de)
Susanne Helbig (susanne.helbig@med.uni-muenchen.de)
Martin Greif (martin.greif@med.uni.muenchen.de)
Janine Ruemmler (janine.ruemmler@med.uni-muenchen.de)
Christoph Becker (christoph.becker@med.uni-muenchen.de)
Marcus Hacker (marcus.hacker@med.uni-muenchen.de)
Gerhard Steinbeck (gerhard.steinbeck@med.uni-muenchen.de)
Alexander Becker (alexander.becker@med.uni-muenchen.de)

Version: 2 Date: 11 May 2012

Author's response to reviews:

Change of description of included patients (Abstract as well as Maintext-Methods-Patients): Included patients had clinical indication for invasive coronary angiography. Additionally correction of typo mistake: hemodynamically significant stenosis was defined as 75% not 70% (although the patient numbers would not change).

Thank you